ACCELERATORS & EXPERIMENTAL FACILITIES STATUS SUMMARY OF WEEK 35 - 2023

Technical infrastructure – C. Pruneaux Linac 4 – P. Skowronski PS Booster – F. Roncarolo ISOLDE – *M. Lozano* PS – R. Garcia Alia PS – East Area – J. Bernhard PS – nTOF – M. Bacak AD – ELENA – L. Bojtar SPS – K. Li SPS – North Area – J. Bernhard SPS – AWAKE – G. Zevi Della Porta SPS - HiRadMat - No report, not running Linac 3 – R. Wegner LEIR – T. Argyropoulos LHC – M. Solfaroli CLEAR - P. Korysko & W. Farabolini

	Technical Infrastructure (TI)										
Facility Cod	ordinator last week	Clément	Pruneaux								
Facility Cod	ordinator this weel	Jesper N	ielsen								
	-		Statistics								
Alarms											
Phone	1	ncoming		Outgoing							
ODMs			1		1						
	1	F	acility Status								
Summary											
Issues	 Mon 28/08/23 10 Emergency stop but (SPS point 4). The p looked like somethin "EUB5/B4" that cut by EN-EL, after wh Wed 30/08/23 11 High level alarm in found to be not work found in a jonction b likely the high vibra bad connections. Wed 30/08/23 13 Major fault on cooli had come off their s in the same state. The full load was pust standby pumping sta cooling tower. During the followin cooling towers. After Thu 31/08/23 09 Electrical perturbati EDF/RTE had not set SIG, however, confi storm. This is most voltage during 90ms 	5:22 ton pushed / versions who a ng hit the butt is the building ich the power 5:48 the pit of the cing. Investig box, inside the tions in this b 6:04 ng tower in S upports and b it on the secon tion to reduc g days EN-CV r which CRY 5:52: on seen in for een anything of rmed that the ikely the cause at CERN.	window broken during an accidentally broke the wint on. The emergency stop of (BB4 + BHA4 (921) + E out be restored. lift in PX15 (ATLAS). O ations on the cabling wer e CRYO building. The ca- building. A campaign is o FA18 (LHC point 18), So broken. After visit on-site andary cooling tower, and e the need for cooling and V managed to do some re O could switch back on b m of multiple alarms bot on their side. y had had a problem with se of the perturbation that	intervention cl adow were not i caused a power 3G4 (930)). The m-site the level re done and a ba use of the bad ngoing to check ome pipes insid it was found th CRYO was ask d to allow work pairs and put ba both compresso h on 48V and U	ose to the lift in BB4 identified, however it cut of the chain e button was replaced was OK, nothing was ad connection was connection is very k all these boxes for e the cooling tower hat several pipes were act to switch off their ting with only one ack in service both or stations to nominal. JPS systems.						

	Mon 04/09/ ALICE dipole has no power building whe	23 02:08 tripped during the nir converter piquet, an re the power convert	ght, ALICE was put in co d no alarms). During the ers are installed tripped to outs are most likely not re	ntact with LHC operations (TI e night the ventilation of the twice, which caused a small elated.
Plans				
		Interven	tion Request	
Yes / No	Duration		Preferred date/time	
Reason				
Impact				

Linac 4									
Machine Coor	dinator last w	veek	Piotr Skowro	onski					
Machine Coor	dinator this v	veek	Jean-Baptist	te Lallement					
	Statistics								
Availability	100.0%								
Facility Status									
Summary	Not a second Past weeks we the beam end this value is se employing er temperature and they are	d of do we fou ergy fl still ful nergy fluctua trying	owntime. Ind that durin luctuates by { lly acceptable painting. We ations. Natura to find a solu	g warm days (above ci 50 keV peak to peak. F e. On the other hand, it prove that this is relate ally, we have communi ution.	irca 22 degrees Celsius) For the operational beams is not for MD beams ed cooling water icated the issue to EN/CV				
Issues	-								
Plans	Keep going v	vith th	is pace.						
			Interventi	on Request					
No	Duration			Preferred date/time					
Reason									
Impact									

PS Booster										
Machine Coor	dinator last week	F.Roncarolo)							
Machine Coor	dinator this week	J-F. Combli	n							
Beam Scheduled										
ISOLDE	Yes		PS	Yes						
	Beam	Availability	by Destination (AFT)							
ISOLDE	98.5 %		PS	98.5%						
		Facilit	y Status							
Summary	 Beam del downtime All was re Planned p on Thu 3² 	 Beam delivered as planned for operation and MDs, with very limited downtime due to issues listed below. All was ready for LHC re-start Planned periodic visual inspection of water leaks (QFO11 and BI.BSW) on Thu 31st completed. Leaks stable. Tue 29th: ~50 min downtime due to BI.DIST10 (distributor) unbalanced 								
Issues	 Tue 29th: measure tl T s Thu 31st: P b T T RF Ring 2 	~50 min down ment hu 31 ^{st,} for pro- hadow of plar BI1.BSW1L1. ower converte umper power- utside PSB bu ook ~1h20m hasking/unma naffected. 2 down for fev	ntime due to BI.DIST10 eased → ok eventive maintenance: ned access 2 trips few times on The er expert intervention converters, which is v uilding suspected to in to repair, during which sking following MP pro-	0 (distributor) unbalanced a changed card in the hu 31 st → found 2 screws below ery unusual → road works duce vibrations. h Ring1 was inhibited (BIC bocedure), R2,3,4						
Plans	 Routine op Next 'routi Wednesda 	peration and M ne access' for ay 13th Septe	/IDs · water leaks visual ins ember	pections: w#37, on						
		Interventi	on Request							
Yes/No	Duration No		Preferred date/time							
Reason										
Impact										

	ISOLDE										
Machine Supe	ervisor last wee	k	Miguel Lo	zano							
Machine Supe	ervisor this wee	k	Alberto R	odriguez							
			Beam S	cheduled							
GPS	Yes	HRS		Stand by	HIE-ISO	Yes					
	Bea	am Av	ailability b	by Destination	(AFT)						
GPS	%	HRS		%	HIE-ISO	83 %					
			Facility	y Status							
Summary	Preparation of the pilot beam setup A/Q=4 at 4.75 MeV/u to Miniball Optimization of the beam injection into Miniball . Proton scan and yield measurements on GPS. Stable beam setup of the low energy part, GPS separator, REXTRAP and REXEBIS using 67Zn21+. Precise beam energy measurements and slow extraction scanned and applied. Radioactive beam, 80Zn21+, delivered to Miniball on Wednesday afternoon one day ahead schedule). Miniball took beam until Saturday night. The experiment had to stop due to a arget failure.										
Issues	-Many trips of t -Four line heati -On Saturday r Miniball. -After some inv was misaligned inside the targe we continued in producing any -During the deg lost with an inc	he RF ng trip nornin restiga J. The et. RIL n this c more { graded rease	Q RF durir s and one g users rep tions it wa electrical r IS realigne degraded r 30Zn and t mode ope of the heat	ng the run. target heating to borted a total loss s concluded that esistance had of d the lasers to the node until 1:30 he experiment le eration we comp ting power goins	rip. st of the 80Zn21+ be to the ion source had change indicating a the new ion source p PM when the target had to stop. bensated the target g into to the line.	eam at d moved and bad contact position, and stopped production					
Plans	On Monday mo Still need to be move to next e	orning decid xperim	the faulty (ed if we wi nent (79Zn	GPS target will I Il continue with at Miniball).	be replaced with a s the 80Zn experimer	pare. ht or if we will					
			Interventi	on Request							
Yes / No	Duration			Preferred o	late/time						
Reason											
Impact											

PS									
Machine Coor	dinator last	week R	uben Garcia Al	ia					
Machine Coordinator this week Denis Cotte									
Beam Scheduled									
East Area	Yes	nTOF	Yes	AD	Yes	SPS	Yes		
	I	Beam Av	ailability by D	estinatio	n (AFT)				
AD	97%	EA N	98%	EA T8	97%	EA T9	98%		
nTOF	98%	SPS	97%						
Facility Status									
Summary	 Success ILHC_# MTE hig Several increase IEAST c accomp Tempor not avai Very go C66 not 	sful comm 3b_75_Pl gh intensit AD TT2 t commissic lished ary use o lable) od availal following	hissioning of ILI b ty MD limited to trajectory check oning ongoing - f 4 bunch AD b bility; no faults g function and to	HC#4b_1(o 3e13 <s and="" rea<br="">- steering beam due above 1h3 riggering I</s>	D0_Pb and (ii adjustments (until end of T to issue in P\$ 30 osses at tran	n case ne e.g. after F8 not yet SB (one o sition on ⁻	eded) intensity f the rings TOF –		
Issues	 required KFA71n radiation setting i PR.MPS by 5ms Not pos STFPRO Issues v 	I cable re n11 pulsir n alarms a t to stand S tripping solved the sible to re O) – circu vith AD q	placement ng somewhere and surpassing by on LHC INDIV e issue estart KFA45 af it breaker had uad which requ	else in the BLM thre due to too iter severa to be rear ired Front	e cycle and tr shold – prob o long cycle - al tries (i.e. no med <u>t Line on-site</u>	iggering r lem disap - shortenii o beam fo <u>interventi</u>	ecurrent peared after ng flat top r AD and ion (twice)		
Plans	 Analysis to study Active n 	s of ion be possible nonitoring	eam lifetime, in impact of BGI of RF gap rela	view of in gas injecti ays in relat	icreased vact ion) tion to recent	uum levels failures	s (and also		
			Intervention R	equest					
No	Duration		Pre	ferred da	te/time				
Reason									
Impact									

PS East Area									
Facility Coordinator last week J. Bernhard									
Facility Coord	linator this	week	L. Nevay						
			Beam Se	cheduled					
T8	Yes	T 9	Yes T10 Yes T11			T11	No		
Beam Availability by Destination (AFT) General:90.8%									
Running T8	97.4%	T 9	97.4%	T10	97.4%		T11	N/A	
Facility Status									
Summarv	T09/T10: Good operation week.								
	T11: No us	er.							
Issues	T10: Brief of	downtime	e from magne	et BHZ027,	solved with	n res	et, no other	issues.	
Plans	T09: HERD T10: MPGI) continution $OCAL \rightarrow$	es. RE7 GAMM	A MEV					
			Interventio	on Request					
Yes	Duration			Preferred of	late/time				
Reason									
Impact									

PS nTOF								
Facility Coord	linator last w	Michael Bac	Michael Bacak					
Facility Coord	linator this w	veek Michael Bac	Michael Bacak					
Beam Requested								
Yes								
Facility Status								
Summary	Progressing EAF dos EAF	with physics progra 1: 12C(n,cp) with s imetry/hadron thera 2: 26Al(n,p) and (amme according t silicon and GEMPi upy. n,a) for Astrophys	o planning. ix detectors for ics				
Issues	No issues o	on experiment side						
Plans	Continue pro	ogramme in EAR1// /access – installatic	2. on of SiC detector	(Wed)				
		Foreseen	Beam Stop					
Yes	Duration	7h	Date/Time	WED 06.09.23; 09h00				

AD - ELENA									
Machine Supe	rvisor last wee	k	Lajos Bojt	Lajos Bojtar					
Machine Supe	rvisor this wee	k	Laurette F	once					
			Beam So	cheduled					
AD	Yes		E	LENA	Yes				
			Availabil	ity (AFT)					
AD	91.3 %		E	LENA	96.4 %				
	Facility Status								
Summary Issues	Good week, no major fault in AD/ELENA. We have record intensity ejected from the AD, above 4E7 pbars. This was mainly due to an increase of the production beam intensity, but also have good deceleration efficiency. MD was done Wednesday on AD coherent oscillations at ejection and AD injection bunch rotation. DR.QUAD went down several times during the week, First line had intervention during Saturday night. Restart of the quad is still problematic. Sunday morning a fast valve in the LNI line (AD to ELENA) closed due to a pressure rise. Vacuum piquet had to be called to open it. It took a bit longer than usual due to phone network issues.								
Plans									
			Interventio	on Request					
Yes / No	Duration			Preferred date	e/time				
Reason									
Impact									

			SP	S					
Machine C	oordinator last	week	Kevin Li						
Machine C	oordinator this	week	Verena Kain						
			Beam S	cheduled					
LHC	Yes N	Α	Yes	AWAKE	Yes	HiRadMat	No		
		Beam	Availability I	oy Destinatio	on (AFT)				
LHC	97.3% N	A	92.7%	AWAKE	95.9%	HiRadMat	-		
			Facilit	y Status					
Summary	An eventful weat at a good 91% LHC coming bac commissioning a dedicated ME Monday and Tu (after a slight in hardware comma able to take beat Tuesday night a Wednesday two blow-up studies operational test INDIV beams in LHC beam delim first trains of 12 hampered filling to be fixed next AWAKE could of running reliably lon beam comma has been done beam looks goor results from las to the RF control is still under inw cycle for the mo	ek for the overall. ick onli has als o on fixe uesday icrease nission am due and ext o dedic s and for s on no the af very co bunch g. This week. only sta over the nission using r od over t year of ol, but t restigat oment.	ne SPS witho On the plan ne from Wed so started in t ed target bea were focused in intensity) ing took place to issues on racted to TEI ated MDs color progressing on-local cryst ternoon; this ntinued all ov es. On the w was traced d art taking bea ne weekend. ing is progress nominals from all. However, could not yet the source of ion. Likely co	but any major was NA and nesday onwa he SPS. In a ms and non- d on delivery as well as run e on Monday the experime Ds. uld successfu yent fixed ta al shadowing went rather s ver the week eekend, BIC own to faulty m late in the ssing with diff n the PS on the slip stacking be successfu the difficultie mmissioning	faults or do AWAKE be ard. Parallel ddition, We local crysta of physics I nning parall afternoon. ents side. L ully be carri arget beams b. LHC start smooth. without ma interlocks of readings o week but h iculties. Ge he short slip y was not ye illy reprodu s has not y will continu	owntimes. Av eam delivery, ion beam ednesday was I shadowing. beams for the lel MDs. Ion the AWAKE was HC beams w ed out for aut s in preparatie ed taking PIL jor issues, tal on the safe be n BCT 4 which as since ther eneral beam so o-stacking cy et achieved a ced. Issues s et been under a with the sh	ailability is with the s booked for e North Area beams not yet ere taken comated on for .OT and king also eam flag ch will need a been setting up cle and the nd the eem linked ort ions		
Issues	Errone	ous pov	wer cut in BB	4	(1)				
Plans	 LHC be pp refe Continu RP rob for 24h 	eam de rence r ue ions ot acce in betv	livery accordi un,) beam comm ess Monday fi veen	ng to LHC pr issioning rom 8-9 and ⁻	ogram (tbd Tuesday 7:	, Van der Me 30-8:30; SFT	er scans, PRO2 run		
			Interventi	on Request					
Yes / No	Duration			Preferred d	ate/time				
Reason									
Impact									

	SPS North Area										
Facility Co	ord	inator la	ast week	J. Bernhard							
Facility Co	ord	inator tl	his week	L. Nevay							
					Beam	Scheduled					
H2	Ye	s	H6	Ye	S	K12	Yes	P42	Yes		
H4	Ye	s	H8	Ye	S	M2	No	TT20	Yes		
		B	eam Availa	bilit	y by Des	tination (AF	T) General:	92.4%			
H2	92.	.8%	H6	92	.8%	K12	92.8%	P42	92.8%		
H4	92.	.8%	H8	91	.9%	M2	91.3%	TT20	92.8%		
Facility Status											
SummaryM2: Slight delay in the installation of the ventilation duct modification for the M tent, completed on 31.08. All beam files tuned and available.P42/K12: Good operation. Sharing: 50 (T2) - 56 (T4) - 50 (T6) Note: T4 always to be adapted to give 22 T10. Already from Monday, 04.09.23: 50 (T2) - 56 (T4) - 100 (T6) due to late request. From Wednesday, 13.09.23, one could reduce T4 to about 37 units, Tilecal stops.					for the MUonE o give 22 on e to late MUonE 37 units, as						
Issues		H2: Mo need to confirm H4: Gol solved. H6: One H8: Issu synchro days. M2: Re physica	ving beam i be identifie ed with SY- liath and Da e power cor ue with patr onising acce ad value of I, minimum	ssue d, S EPC avid i nvert olling ess ri XCN is –2	e still pres Y-EPC fo availabil magnet p er fault ca g PPE14 ghts from /IH.06175 25 mm). F	sent due to m ollowing. Tes lity. ower conver aused 2 h do 3, access do 5 the databas 52 muon scra Piquet fixed t	nagnet / pow ts planned fo ters take too owntime. or unit restar se, will be ch aper went to the reading,	er converter or next week o long to ram rted as it wa necked agair –6 mm on 2 45 min down	r problems that s, to be p up, issue s not n in the coming 28.08. (not ntime.		
Plans		H2: L H4: F H6: M H8: A M2: N	$HCb \rightarrow NA$ RD51/GIF+- MONOLITH ATLAS Tile(MUonE cont)	.65 (⊦ cor , EP Cal c inue	DsTau) ntinues. PIXEL — continues, s.	→ ATLAS HG , LHCb+SND	TD, AIDAIN	NOVA, RD4 arasitically.	2.		
					Intervent	ion Reques	ts				
No		Duratio	on		Preferre	ed date/time	•				

SPS AWAKE											
Facility Coord	inator last w	eek Giovanni Ze	evi Della I	Porta							
Facility Coord	inator this w	/eek -									
		Facilit	y Status								
Summary	Summary of Most of the y caused unst laser timing, Saturday mo electron line Tuesday/We Other activit Insta Elec Plas Proton beam Sho pre- Data stuc	of Laser issue: week spent unders able/unusable plas and proton+plasm protocathode), so ednesday to bring b ies in the shadow of alled additional µs of ctron beam steering sma light PMT calib n: rt tests (~200 extra pulse a taking on Saturda dies ctions eam to AWAKE out beam (including p changes)	tanding a ma condi a experin ly, the cu additiona back the e of laser ac camera fo g tests ration ctions) or by and Su M 139 0	nd so tions. nents rrent s al work electro ccess: or plas n Mon nday:	Iving a The pr have b solutior k is pla on bear sma ligi day an densit	laser problem een provennned on nned on nt d Thur y step Th 279 2.6 0	rsday	ulse issu solved b e since JV (need the MD the MD blasma li blasma li S 1390 8.7 4.8	e which by tuning led for s of c laser ght S 1073 6.6 2.1		
Issues Plans	Lase Vap inve it. C Elec by E SPS solv Proton run: :	er pre-pulse issue s or source OTC circ estigation. Another (contractors interven ctron line power sup EPC calibration. S: another RBI.4100 red by Piquet. study the effect of a	solved on uit failure OTC failu tion planr oply (RCII 010 issue	Frida bypa re on ned du BH.41 on Th densi	ty after issed b Sunda uring ne (2349) hursday	a full v ut not y: 1.5 ext We high jit y (repe using	veek unde hour a dnes ter so eat of plasn	of invest rstood, u access t day MD. olved on last Sun na light a	rigation Inder o patch Monday day),		
		Foreseer	beam s	op							
Yes / No	Duration		date/tin	1e							

LHC						
Machine Coordinator last week		M.Solfaroli				
Machine Coordinator this week		E.Metral				
Statistics						
Availability	55.1%		Stable Beam Ratio 3.2%			
Facility Status						
Summary	Facility StatusPowering tests completed on Wednesday morning. RQX.L8 and RD1.L8 commissioned to nominal current without quench. Long investigation for unusual interlock propagation of RCBXV1.L8.Problem identified on the CIPA PIC unit. After replacement, PIC tests repeated for all circuits in XL8. Beam operation restored on Wednesday afternoon. Beams directly circulating. Horizontal orbit perturbation in IP8 found much smaller than expected. Measured optics (inj, 1.2m, 60cm, 30cm) and aperture (inj and after LHCb rotation), B2 injection setup and performed loss maps at injection on pp nominal cycle. All results are in line with the measurements taken in April. Impressive reproducibility.Four hours of stable beams at injection on Friday for detector recommissioning. All different configurations tested, cleaned and prepared:• intermediate energy run (up to TCT/XRP alignment) • VdM (TCT/XRP alignment + LMs)• 3/6 km high beta (optics measured and cycle cleaned, ready for background test)IP8-TDIS vacuum degraded between Thursday and Friday evening. Investigation (on saturday) revealed a leak on module B, right (bottom) jaw. As small intensity operation was foreseen for the weekend, the jaw was condemned (driver blocked by CEM) and the interlock limits adjusted. Impact for ppref and IONS runs to be evaluated to decide on possible exchange (spare being prepared).Two trips of RB.A78 on Earth_fault (4 events this year). Visul inspection did not reveal any evident issue. Few pending access for RQ4.L8 PC, BLM in IP6 (XPOC), 60A and RSS converter.TDI-IP8 vacuum (module B, bottom jaw) RB.A78 earth_fault (intermittent)Very high beta (background test and run) Van der Meer run Finalzation of pp ref commissioning					
Issues	TDI-IP8 vacuum (r RB.A78 earth faul	TDI-IP8 vacuum (module B, bottom jaw) BB A78 earth_fault (intermittent)				
Plans	Very high beta (background test and run) Van der Meer run Finalization of pp ref commissioning					
Intervention Request						
Yes / No	Duration		Preferred date/time			

Linac 3				
Machine Supervisor last week		R. Wegner		
Machine Supervisor this week		G. Bellodi		
Statistics				
Availability	98.4%			
Facility Status				
lon species	lead			
Summary	 Monday oven refilled, quick restart, NO vacuum issue From Tuesday afternoon mostly stable beam >35 mA out of Linac3 RF remote reset capabilities extended digital LLRF timings included to timing working set (see issue below), cycle cloning is working now 			
Issues	 some digital LLRF timings not propagated from FESA to LSA => cloned cycle did not accelerate beam Restart issues of digital LLRF systems (RFQ, Tank1, Buncher). Expert informed. Work-around: contact RF amplifier specialist who repeats restart until digital LLRF works. Buncher: Same issue + phase setpoint might need to be re-adjusted by a multiple of 45 deg, depending on LLRF phase locking. 			
Plans	continue stable b	continue stable beam production		
Intervention Request				
Yes / No	Duration	Preferred date/time		
Reason				
Impact				

LEIR					
Machine Supervisor last week		Theodoros A	Theodoros Argyropoulos		
Machine Supervisor this week		Theodoros A	Theodoros Argyropoulos		
Statistics					
Availability	Beam Commissioning				
Facility Status					
lon species	Pb				
Summary	 Commissioning of NOMINAL beam continued. Good injection efficiency (~40-50%) achieved for almost all 7 injections. Issue with the accumulation of the beam. No efficient cooling of some parts of the beam in the longitudinal plane. Investigations of the e-cooler and other possible issues (LINAC3 settings, beam sizes etc.). No clear conclusion yet. The 75ns beam was tested and extracted to the PS. 				
Issues					
Plans	-Continue commissioning of NOMINAL cycle.				
Intervention Request					
Yes / No	Duration		Preferred date/time		
Reason					
Impact					

CLEAR				
Facility Coordinator last week		Pierre Korysko & Wilfrid Farabolini		
Facility Coordinator this week		Pierre Korysko		
Facility Status				
Summary	Last week was dedicated to two experiments: - Uniform beam irradiations using a double-scattering foil system (with the University of Oxford). - Dosimetry studies for Cancer Therapy with VHEE at UHDR.			
Issues	No major issue.			
Plans	This week is dedicated to two experiments: - AWAKE ChDR BPM measurements (with Univ. of Oxford and CERN-BI). - Chemistry studies with Very High Energy Electrons (VHEE) at Ultra High Dose Rate (UHDR) to observe the FLASH Effect (with CHUV).			